Author’s response to reviews

Title: Work related characteristics, work-home and home-work interference and burnout among primary healthcare physicians: A gender perspective in a Serbian context

Authors:

Katarina Putnik (k.putnik@maastrichtuniversity.nl)
Inge Houkes (inge.houkes@maastrichtuniversity.nl)

Version: 4 Date: 19 September 2011

Author’s response to reviews:

Dear Editorial Production Team of BMC Public Health,

Please find below the comments we have received from you and how we have addressed them:

Major revisions:

1. Abstract: This should be identical in the manuscript and the submission system

Action taken: We will make sure when uploading the changes that the abstract in the manuscript and in the submission system is the same

2. Structure: Please include conclusion, after discussion.

Action taken: We have changed the structure, so that the following is the sequence of the final headings: first is the Discussion, then Recommendations for future research, followed by Recommendations for policy and practice, and lastly the Conclusion paragraph.

3. The change from the previous version is that we have put Recommendations for future research and policy and practice before the Conclusion section (rather than after it).

4. Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

Action taken: We have written the following sentence: KP and IH drafted and revised the manuscript and gave approval for the final version of the manuscript to be published.
5. References: Please provide functioning link to


Action taken: The link has been adjusted (spelling mistake was discovered). The link now reads:


6. Additional files: Please remove from the manuscript.

Action taken: We did not have additional files to the manuscript

Minor revisions

1. Highlighting/tracking: Please remove from the manuscript.

Action taken: Tracking removed from the manuscript

2. Please remove 'Burnout in Serbia' from each page.

Action taken: The title “Burnout in Serbia” is removed from each page

3. Comments: Please remove post-it style comments. Eg: Table 1 here.

Action taken: Post-it comments removed throughout the manuscript

4. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Action taken: We checked the manuscript for typographical mistakes.

Should you require any other changes, or if there are any uncertainties, please feel free to contact us.

Thank you.

Sincerely yours,